Author’s response to reviews

Title: Improving Motivation and Retention of Health Professionals in Developing Countries: A Systematic Review

Authors:

Mischa Willis (willismm@tcd.ie)
Posy Bidwell (bidwellp@tcd.ie)
Stephen Thomas (steve.thomas@tcd.ie)
Laura Wyness (LWyness@whc.ie)
Duane Blaauw (duane.blaauw@nhls.ac.za)
Prudence Ditlopo (prudence.ditlopo@nhls.ac.za)

Version: 2 Date: 5 June 2008

Author’s response to reviews:

Mark Todd
Assistant Editor

Dear Mark,

In accordance with your recent email, we have attached the title page and abstract to the main manuscript. We would also like to continue with our application to BMC HSR rather than BMC IHHR, because the BMC HSR has published similar work and has a larger audience and we believe our research topic has relevance beyond LMIC countries.

Many thanks for your consideration.

Kind Regards

Steve Thomas